
SOHO Daily Meeting Minutes for 20 May 1999

DOY: 140

FOT Report
----------
Spacecraft Status:    Nominal
Spacecraft Anomalies: None

Accomplished Activities: VIRGO, SWAN, Extra 2 hrs VC2

Planned Activities: VIRGO, Nominal HGA Table, Extra 2 hrs VC2

Upcoming Operations: 19-21 May - Extra 2 hrs VC2 daily

Ground Anomalies: 
  139/1437 - CPA abort at D24; Command ability lost (11 min)

SOHO daily meeting at MEDOC
---------------------------
All closed doors where reopened.
      
SUMER switched to detector B which works normally and will need 
some trimming in order to deal with intense lines.
Good spectra for both JOPs  Detailed analysis will be possible.
      
CDS identified a bright point associated with a plume in the CH 
approaching sun center.  Features identified in line ratios 
sensitive to temperature (hot BP cool Plume) and FIP effect.  
JOP 102 should work.
      
EIT identified a monster CME between 1800 UT and 1900 UT due to 
apparently temporarily corrupted data in the QKL data base. The 
fever has dropped.
      
Some small ejecta all day from AR 8545
Small filament eruption in the NW at  2000 UT
      
This morning fair size CME from AR 8545 producing noticeable 
dimming at 0900 UT.

Pic du Midi was in severe weather condition : snow and no 
helicopter.  

Fair weather expected at the Pic this weekend ( no EIT !)
      
JOPs in preparation for tomorrow:
    JOP  102 target  around 260 ,  -20
    JOP  101 target 0 , 0
    JOP  38  target  around +180 , +5  EIT contribution requested
                 
    Expecting Inputs from Shaddia Habbal ASAP
       
       
                                               JPD (SOL)


